Author's response to reviews

Title: Polymorphism of the FABP2 gene: a population frequency analysis and an association study with cardiovascular risk markers in Argentina.

Authors:

Laura L C Gomez (gomez@fcm.uncu.edu.ar)
Sebastian S M Real (sreal@unsl.edu.ar)
Marta M S Ojeda (msojeda@unsl.edu.ar)
Sergio S Gimenez (sergio.gimenez@osep.mendoza.gov.ar)
Luis L S Mayorga (lmayorga@fcm.uncu.edu.ar)
Maria M Roque (mroque@fcm.uncu.edu.ar)

Version: 2 Date: 30 January 2007

Author's response to reviews:

The new revision of our manuscript contains little changes in details like deleted dots at the end of sentences and incomplete page number in a Reference.